

# Association of body mass index and cardiotoxicity related to anthracyclines and trastuzumab in early breast cancer: French CANTO cohort study

Elisé G Kaboré, Charles Guenancia, Ines Vaz-Luis, Antonio Di Meglio, Barbara Pistilli, Charles Coutant, Paul Cottu, Anne Lesur, Thierry Petit,

Florence Dalenc, et al.

# ▶ To cite this version:

Elisé G Kaboré, Charles Guenancia, Ines Vaz-Luis, Antonio Di Meglio, Barbara Pistilli, et al.. Association of body mass index and cardiotoxicity related to anthracyclines and trastuzumab in early breast cancer: French CANTO cohort study. PLoS Medicine, 2019, 16 (12), pp.e1002989. 10.1371/journal.pmed.1002989. inserm-02430330

# HAL Id: inserm-02430330 https://inserm.hal.science/inserm-02430330v1

Submitted on 7 Jan 2020

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# 

**Citation:** Kaboré EG, Guenancia C, Vaz-Luis I, Di Meglio A, Pistilli B, Coutant C, et al. (2019) Association of body mass index and cardiotoxicity related to anthracyclines and trastuzumab in early breast cancer: French CANTO cohort study. PLoS Med 16(12): e1002989. https://doi.org/10.1371/ journal.pmed.1002989

Academic Editor: Sanjay Basu, Harvard Medical School, UNITED STATES

Received: July 28, 2019

Accepted: November 15, 2019

Published: December 23, 2019

**Copyright:** © 2019 Kaboré et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data access is restricted due to legal concern as it is third party data. Data are available to all interested researchers upon request to the 3C study ethics committee. A written request should be submitted to CANTO scientific committee (s-everhard@unicancer.fr)

**Funding:** This research was supported by the French Government under the "Investment for the Future" program managed by the National Research Agency (ANR), grant n° ANR-10-COHO- RESEARCH ARTICLE

# Association of body mass index and cardiotoxicity related to anthracyclines and trastuzumab in early breast cancer: French CANTO cohort study

Elisé G. Kaboré<sup>1®</sup>, Charles Guenancia<sup>2®</sup>\*, Ines Vaz-Luis<sup>3</sup>, Antonio Di Meglio<sup>3</sup>, Barbara Pistilli<sup>3</sup>, Charles Coutant<sup>4</sup>, Paul Cottu<sup>5</sup>, Anne Lesur<sup>6</sup>, Thierry Petit<sup>7</sup>, Florence Dalenc<sup>8</sup>, Philippe Rouanet<sup>9</sup>, Antoine Arnaud<sup>10</sup>, Olivier Arsene<sup>11</sup>, Mahmoud Ibrahim<sup>12</sup>, Johanna Wassermann<sup>13</sup>, Geneviève Boileau-Jolimoy<sup>14</sup>, Anne-Laure Martin<sup>15</sup>, Jérôme Lemonnier<sup>15</sup>, Fabrice André<sup>3</sup>, Patrick Arveux<sup>11,4</sup>

 "Health across Generations" Team, Inserm U1018, Centre for Research in Epidemiology and Population Health (CESP), Villejuif, France, 2 Cardiology Department, University Hospital Dijon Bourgogne, Dijon, France, 3 Institut Gustave Roussy, Villejuif, France, 4 Centre Georges-François Leclerc, Dijon, France, 5 Institut Curie, Paris, France, 6 Centre Alexis Vautrin, Vandoeuvre les Nancy, France, 7 Centre Paul Strauss, Strasbourg, France, 8 Institut Claudius Regaud, Toulouse, France, 9 Institut du Cancer de Montpellier, Montpellier, France, 10 Clinique Sainte Catherine, Avignon, France, 11 Centre Hospitalier de Blois, Blois, France, 12 Centre Hospitalier Régional d'Orléans, Orléans, France, 13 AP-HP Pitié-Salpêtrière, Paris, France, 14 Polyclinique du Parc Drevon, Dijon, France, 15 Unicancer, Paris, France

# Abstract

# Background

In patients treated with cardiotoxic chemotherapies, the presence of cardiovascular risk factors and previous cardiac disease have been strongly correlated to the onset of cardiotoxicity. The influence of overweight and obesity as risk factors in the development of treatment-related cardiotoxicity in breast cancer (BC) was recently suggested. However, due to meta-analysis design, it was not possible to take into account associated cardiac risk factors or other classic risk factors for anthracycline (antineoplastic antibiotic) and trastuzumab (monoclonal antibody) cardiotoxicity.

# Methods and findings

Using prospective data collected from 2012–2014 in the French national multicenter prospective CANTO (CANcer TOxicities) study of 26 French cancer centers, we aimed to examine the association of body mass index (BMI) and cardiotoxicity (defined as a reduction in left ventricular ejection fraction [LVEF] > 10 percentage points from baseline to LVEF < 50%). In total, 929 patients with stage I–III BC (mean age  $52 \pm 11$  years, mean BMI  $25.6 \pm 5.1$  kg/m<sup>2</sup>, 42% with 1 or more cardiovascular risk factors) treated with anthracycline (86% epirubicin, 7% doxorubicin) and/or trastuzumab (36%), with LVEF measurement at baseline and at least 1 assessment post-chemotherapy were eligible in this interim analysis. We analyzed associations between BMI and cardiotoxicity using multivariate logistic regression. At

These authors contributed equally to this work.
\* charles.guenancia@chu-dijon.fr



0004 (FA) and by a Clinical Research Fellowship from the French government (EGK). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: IVL declares honoraria from AstraZeneca, Kephren and Novartis, outside the submitted work. BP declares grants from Pierre-Fabre, personal fees from AstraZeneca and Pfizer and non-financial support from Pfizer, Puma and Merus, outside the submitted work.

Abbreviations: BC, breast cancer; BMI, body mass index; CVD, cardiovascular disease; LVEF, left ventricular ejection fraction; MUGA, multigated acquisition; OR, odds ratio. baseline, nearly 50% of the study population was overweight or obese. During a mean follow-up of 22 ± 2 months following treatment completion, cardiotoxicity occurred in 29 patients (3.2%). The obese group was more prone to cardiotoxicity than the normal-weight group (9/ 171 versus 8/466; p = 0.01). In multivariate analysis, obesity (odds ratio [OR] 3.02; 95% CI 1.10–8.25; p = 0.03) and administration of trastuzumab (OR 12.12; 95% CI 3.6–40.4; p <0.001) were independently associated with cardiotoxicity. Selection bias and relatively short follow-up are potential limitations of this national multicenter observational cohort.

## Conclusions

In BC patients, obesity appears to be associated with an important increase in risk-related cardiotoxicity (CANTO, ClinicalTrials.gov registry ID: NCT01993498).

#### **Trial registration**

ClinicalTrials.gov NCT01993498.

### Author summary

#### Why was this study done?

- Anthracyclines remain a cornerstone of breast cancer therapy, in combination with new-generation targeted drugs such as trastuzumab, and represent the major culprit in chemotherapy-induced heart disease.
- In a recent meta-analysis, we showed that overweight and obesity increased cardiotoxicity, but it was not possible to take into account associated cardiac risk factors or other classic risk factors for anthracycline and trastuzumab cardiotoxicity.
- We examined the association of body mass index (BMI) and cardiotoxicity using prospective data collected in the CANTO study, a French national cohort.

#### What did the researchers do and find?

- In patients treated for stage I–III breast cancer with anthracyclines and/or trastuzumab, being obese was associated with increased risk of developing cardiotoxicity, regardless of other predictors of cardiotoxicity.
- Among all the classic risk factors for cardiotoxicity, overweight and obesity were by far the most common risk factors in this breast cancer population.

## What do these findings mean?

- Obesity appears to be a risk factor for cardiotoxicity in breast cancer patients.
- Overweight and obese patients may benefit from careful cardiac screening and followup during and after chemotherapy.

## Introduction

Remarkable progress in the treatment of early breast cancer (stages I–IIIA), including multiple combinations of drugs, radiation therapy, and surgery, has been achieved in the past 2 decades. Nevertheless, anthracyclines remain a key element of breast cancer (BC) therapy in combination with new-generation targeted drugs such as trastuzumab, and represent an important cause of chemotherapy-induced heart disease [1].

Cardiotoxicity is a serious side effect of both agents, and its onset may even occur months to years after completion of cancer primary treatment [2,3]. Cardiotoxicity may severely impair the quality of life and overall survival of BC patients [3].

Cancer and cardiovascular disease (CVD) were previously considered 2 different pathologies. Recent data show that they share multiple risk factors, suggesting that there might be a common biological pathway [4]. A high body mass index (BMI) at diagnosis is often associated with a worse prognosis in BC [5,6]. Epidemiological studies show that obesity can increase the incidence of some BCs, lead to a poorer treatment outcome and quality of life after cancer diagnosis, and increase cancer-related mortality [7,8]. Recently, the influence of overweight and obesity as aggravating factors in the development of cardiotoxicity has been highlighted [9]. Animal models have suggested that overweight and obesity increase the risk of cardiotoxicity [10,11].

In a recent meta-analysis, we showed that overweight and obesity were risk factors for cardiotoxicity in treatment with anthracyclines and sequential anthracyclines and trastuzumab [9]. However, due to meta-analysis design, we could not consider associated cardiac risk factors or other classic risk factors for anthracycline and trastuzumab cardiotoxicity (older age, concomitant chemotherapy or previous radiation therapy, and having multiple cardiovascular risk factors such as smoking, hypertension, diabetes, and dyslipidemia).

Thus, using prospective data collected in the CANTO study, a French national cohort, we aimed to examine the association of BMI and cardiotoxicity.

### Methods

#### Study design

OCATO (Obesity and CArdioTOxicity in breast cancer) was an ancillary study to the CANTO (CANcer TOxicities) trial [12]. CANTO is a French longitudinal multicenter cohort study designed to evaluate chronic toxicities in patients treated for non-metastatic BC. Patient recruitment and follow-up (from March 2012 to December 2014) was carried out in 26 French cancer centers. Patients were systematically assessed at diagnosis and at 3–6, 12, 36, 48, and 60 months after treatment completion, defined as end of chemotherapy, radiotherapy, or surgery. Trastuzumab and endocrine therapy could be ongoing. This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guidelines (S1 Checklist). This study is supported by the Unicancer group. All patients enrolled in the study provided written informed consent. Approval was obtained from French authorities (Agence française de sécurité sanitaire et des produits de santé: B111158-20) and ethics committees (Comité de protection des personnes: 11–039). An independent international advisory board was set up to guarantee the integrity of the study (ClinicalTrials.gov registry ID: NCT01993498; ID-RCB: 2011-A01095-36).

#### **Study population**

CANTO patients met the following inclusion criteria: female patients aged 18 years and over covered by the national social security system, with histologically proven non-metastatic invasive BC (cT0 to cT3, CN0–3) and without previous cancer treatment.



Fig 1. Flowchart. ANTH, anthracycline; LVEF, left ventricular ejection fraction; M0, month 0; TRZ, trastuzumab. https://doi.org/10.1371/journal.pmed.1002989.g001

We accessed information at baseline, 3–6 months, and 12 months after the end of treatment from 5,801 women with data available in the most recent CANTO data lock (July 2018). For this interim analysis, we excluded patients treated with different therapies other than standard chemotherapy including anthracycline and/or trastuzumab (n = 2,929), patients with no left ventricular ejection fraction (LVEF) measurement at baseline (n = 918) or no available followup measurements (n = 799), patients with LVEF at baseline < 50% (n = 13), patients with BMI  $< 18.5 \text{ kg/m}^2$  (n = 32), and patients with evidence of local, regional, or metastatic recurrence of BC (n = 181). Our final analytic cohort included 929 patients (Fig 1).

#### Study outcomes

In the present study, the primary end point was occurrence of cardiotoxicity. Cardiotoxicity was defined in accordance with current guidelines as a reduction in LVEF > 10 percentage points from baseline to LVEF < 50% [13].

#### Study covariates

Our independent variable was BMI at baseline, categorized following WHO as overweight ( $25 \le BMI < 30 \text{ kg/m}^2$ ), obese (BMI  $\ge 30 \text{ kg/m}^2$ ), and normal ( $18.5 \le BMI < 25 \text{ kg/m}^2$ ) [14].

As covariates we included demographic, clinical, and treatment variables, such as age at diagnosis; cardiovascular risk factors collected at baseline including previous hypertension, diabetes mellitus, dyslipidemia, and current smoking; dose of anthracycline; type of chemo-therapy; and left-sided radiotherapy performed.

#### Statistical analysis

Normal distribution of variables was verified using the Kolmogorov–Smirnov test. Normally distributed continuous data were expressed as mean values ( $\pm$ SD). Differences between groups were analyzed for statistical significance with the *t* test,  $\chi^2$  test, or Fisher exact test as appropriate. Categorical variables were reported as percentages.

A multivariate regression analysis was used to explore the relationships between cardiotoxicity and BMI and baseline covariates. For this analysis, a forward stepwise procedure was used in the logistic regression to select significant factors, in which  $\alpha$  (in) = 0.20 and  $\alpha$  (out) = 0.10. A second multivariate logistic regression analysis was carried out after adjustment for known cardiovascular risk factors [15]. The odds ratios (ORs) and 95% confidence intervals (CIs) were estimated from regression coefficients. A *p*-value of <0.05 was considered statistically significant. All analyses were performed using the SAS statistical package (version 9.4; SAS Institute, Cary, NC).

#### Results

We identified a total of 929 patients who met our inclusion criteria. The anthracycline regimens administered to patients included epirubicin (86%) and doxorubicin (7%). Trastuzumab was administered alone to 67 patients (7%) and combined with standard chemotherapy including anthracycline for 338 patients (36%).

Patients were predominantly younger than 65 years (87%), with a mean age of  $52 \pm 11$  years (Table 1). In all, 394 (42%) patients had 1 or more cardiovascular risk factors. At baseline, nearly 50% of the study population was overweight or obese.

Higher BMI was associated with a higher prevalence of cardiovascular risk factors (Fig 2). Overweight and obese patients were more likely to have previous hypertension (p < 0.001) and diabetes (p < 0.001).

During a mean follow-up of  $22 \pm 2$  months following treatment completion, cardiotoxicity occurred in 29 patients (3.2%), of whom 23 (73%) received anthracycline and trastuzumab, 5 (17%) received trastuzumab alone, and 3 (10%) received anthracycline alone. The preferred tool for cardiac function assessment was nuclear cardiac imaging by multigated acquisition (MUGA) (77%), with no difference regarding the diagnostic tool (MUGA or echocardiography) between the patients who developed cardiotoxicity and those who did not (Table 1).

BMI category distribution (p = 0.01) and mean age (p < 0.001) at baseline were significantly different between patients with and without cardiotoxicity. Compared with individuals in the normal-weight group (8/466), individuals in the overweight (12/292) and obese (9/171) groups were more likely to have cardiotoxicity (p = 0.01) (Fig 3). There was no significant difference in cardiotoxicity occurrence between overweight and obese group (p = 0.56).

In multivariate analysis, BMI  $\geq$  30 kg/m<sup>2</sup> and trastuzumab administration were independently associated with cardiotoxicity (Table 2). When BMI category was included in a multivariate analysis adjusted for having multiple cardiovascular risk factors ( $\geq$ 2 risk factors among smoking, hypertension, diabetes, and dyslipidemia) and trastuzumab administration, obesity (OR 3.02; 95% CI 1.10–8.25; *p* = 0.03) but not overweight (OR 2.43; 95% CI 0.97–6.11) was independently associated with cardiotoxicity (Table 3).

#### Discussion

In this large-scale prospective study, we found that in BC patients treated with anthracycline and/or trastuzumab, being obese was independently associated with a higher rate of cardio-toxicity compared to normal-weight patients.

#### Table 1. Baseline characteristics.

Variable	Total (N = 929)	Cardiotoxicity ( $n = 29$ )	No cardiotoxicity ( $n = 900$ )	p-Value
Age (years)	$52.4 \pm 11.2$	$56.3 \pm 10.9$	$52.2 \pm 11.1$	< 0.001
Age $> 65$ years	125 (13.4%)	7 (24.1%)	118 (13.1%)	0.08
Age $< 40$ years	110 (11.8%)	2 (6.8%)	108 (12.0%)	0.56*
BMI (kg/m <sup>2</sup> )	25.6 ± 5.1	27.51 ± 5.3	25.6 ± 5.1	< 0.001
$BMI \geq 25 \text{ kg/m}^2$	463 (49.8%)	21 (72.4%)	442 (49.1%)	0.01
Cardiovascular risk factors				
Hypertension	187 (20.1%)	9 (31.0%)	178 (19.8%)	0.13
Diabetes mellitus	39 (4.2%)	2 (6.9%)	37 (4.1%)	0.34*
Dyslipidemia	92 (9.9%)	6 (20.7%)	86 (9.5%)	0.04
Smoking	172 (18.5%)	7 (24.1%)	165 (18.3%)	0.42
Coronary artery disease	2 (0.2%)	0 (0%)	2 (0.2%)	_
High cumulative dose of anthracycline				
Doxorubicin $> 250 \text{ mg/m}^2$	3 (0.3%)	0 (0%)	3 (0.3%)	_
Epirubicin > 600 mg/m <sup>2</sup>	0 (0%)	0 (0%)	0 (0%)	_
Chemotherapy agents				
Doxorubicin	62 (6.6%)	2 (6.9%)	60 (6.7%)	0.99*
Epirubicin	800 (86.1%)	22 (75.9%)	778 (86.4%)	0.10
Trastuzumab	405 (43.6%)	26 (89.7%)	379 (42.1%)	< 0.001
Taxane	890 (95.8%)	28 (96.5%)	862 (95.8%)	0.99*
Alkylating agent	867 (93.3%)	28 (96.5%)	839 (93.2%)	0.71*
Left chest wall radiotherapy	484 (51.8%)	13 (44.8%)	471 (52.3%)	0.42
LVEF (%)	$65.8 \pm 6.3$	$61.2 \pm 5.8$	$65.9 \pm 6.2$	< 0.001
Diagnostic tool				0.72
Nuclear cardiac imaging (MUGA)	713 (76.9%)	20 (74.1%)	693 (77.0%)	
Echocardiography	214 (23.1%)	7 (25.9%)	207 (23.0%)	
Follow-up (months)	$22.5 \pm 2.2$	21.3 ± 2.7	22.6 ± 2.2	<0.001

Data given as mean  $\pm$  SD or number (percent). Missing data (*n*): diagnostic tool (2).

\*Fisher exact test.

BMI, body mass index; LVEF, left ventricular ejection fraction; MUGA, multigated acquisition.

https://doi.org/10.1371/journal.pmed.1002989.t001

In the literature, clinical decompensation occurs in 2%-4%, subclinical structural change in 9%-11%, and biomarker rise in 30%-35% of cancer chemotherapy patients [16], mainly depending on the total dose of anthracycline and on the criteria employed to define cardiotoxicity [17]. In our study, the incidence of cardiotoxicity was 3.2%. This low rate of cardiotoxicity could be explained by the definition used. Indeed, until recently, cardiotoxicity definitions were heterogeneous among the various studies published. The cardiac side effects of anthracycline and trastuzumab are mainly expressed by a decrease in LVEF [13]. Different cutoff values of LVEF have been previously used to define cardiac toxicity [18]. We followed a European Society of Cardiology position paper on cancer treatments and cardiovascular toxicity that defined cardiotoxicity as a drop of LVEF of at least 10 points to a LVEF < 50% [13]. Moreover, the CANTO study is in a modern setting of chemotherapy and mainly includes a young and healthy BC population. Considering that the classic risk factors of cardiotoxicity are older age, high anthracycline cumulative doses, and previous CVD, the relatively low rate of cardiotoxicity can thus be easily explained.

The incidence of cardiotoxicity after chemotherapy agents may be increased by smoking or preexisting CVD (or individual patient genetic predisposition to CVD) [13,19,20]. Data





Fig 2. Distribution of cardiovascular risk factors according to body mass index category. \*p < 0.05 versus normal weight; †p < 0.05 versus overweight.

https://doi.org/10.1371/journal.pmed.1002989.g002



https://doi.org/10.1371/journal.pmed.1002989.g003

	MEDICINE
--	----------

Characteristics	Univariate analysis			Multivariate analysis		
	OR	95% CI	p-Value	OR	95% CI	p-Value
Age > 65 years	2.10	0.88-5.04	0.09			
BMI: overweight versus normal	2.45	0.99-6.07	0.05	2.47	0.99-6.20	0.05
BMI: obese versus normal	3.18	1.20-8.38	0.01	3.35	1.25-8.97	0.01
Hypertension	1.82	0.81-4.07	0.14			
Diabetes mellitus	1.72	0.39-7.54	0.46			
Dyslipidemia	2.46	0.97-6.23	0.05			
Smoking	1.41	0.59-3.37	0.43			
High cumulative dose of anthracycline	_	_	_			
Trastuzumab	11.91	3.58-39.65	< 0.001	12.12	3.6-40.4	< 0.001
Taxane	1.23	0.16-9.31	0.83			
Alkylating agent	2.03	0.27-15.21	0.48			
Left chest wall radiotherapy	0.74	0.35-1.55	0.42			

Table 2. Univariate and multivariate analysis of the factors independently associated with cardiotoxicity.

BMI, body mass index; OR, odds ratio.

https://doi.org/10.1371/journal.pmed.1002989.t002

obtained in large cohorts show that patients who have survived cancer are significantly more prone to have cardiovascular risk factors such as overweight, hypertension, diabetes, and dyslipidemia compared with matched controls who have not had cancer [15]. Higher risk of CVD and CVD-related death in cancer survivors is probably due to multiple factors. It might involve cardiotoxic effects of different cancer treatments as well as comorbidities and unhealthy life-style habits [19].

In our study, cardiotoxicity was associated with cardiovascular risk factors such as age, dyslipidemia, and overweight/obesity. Interestingly, despite the absence of previously known CVDs, patients who developed cardiotoxicity had a lower (but normal) LVEF at the beginning of the chemotherapy. These results highlight the higher prevalence of cardiovascular risk factors in these patients and the need to carefully assess cardiological background before any cardiotoxic drug prescription. After multivariate analysis, among all cardiovascular risk factors, only obesity remained significantly associated with cardiotoxicity.

We found in this cohort that being obese was a major and an independent risk factor for cardiotoxicity. The epidemiological association between obesity and heart failure is well established. Kenchaiah et al. [21,22] described in 2 studies that every 1-kg/m<sup>2</sup> increase in BMI was associated with an 11% increased risk of heart failure. Moreover, in these studies, overweight and obese patients had respectively a 49% and a 180% increased risk of developing heart failure

Table 3. Multivariate analysis of the association between BMI category and cardiotoxicity.

Model	OR (95% CI)	OR (95% CI)				
	Normal weight	Overweight	Obese			
Model 1	1 (Reference)	2.45 (0.99-6.07)	3.18 (1.20-8.38)			
Model 2	1 (Reference)	2.43 (0.98-6.03)	2.87 (1.07-7.74)			
Model 3	1 (Reference)	2.43 (0.97-6.11)	3.02 (1.10-8.25)			

Model performance (area under curve [AUC], Hosmer and Lemeshow goodness-of-fit test): Model 1: BMI category; AUC 0.64 (95% CI 0.56–0.74),  $\chi^2$  2.24, p = 0.69. Model 2: Model 1 + multiple cardiovascular risk factors ( $\geq 2$  risk factors among smoking, hypertension, diabetes, and dyslipidemia); AUC 0.64 (95% CI 0.54–0.74),  $\chi^2$  0.17, p = 0.98. Model 3: Model 2 + trastuzumab; AUC 0.80 (95% CI 0.74–0.86),  $\chi^2$  2.04, p = 0.84.

https://doi.org/10.1371/journal.pmed.1002989.t003

at follow-up. In heart failure patients, obesity is an important clinical factor in both acute and long-term care. Obesity was associated with higher acute severity of heart failure in a dataset of 219,465 patients hospitalized for heart failure exacerbation [23].

Emerging evidence indicates a link between obesity and cardiotoxicity. We recently demonstrated in a rodent model submitted to postnatal overfeeding that overweight was associated with more severe cardiac alterations following the administration of anthracyclines and trastuzumab [11]. Therefore, we tested whether these results could be transposed to humans. We performed a random-effects and network meta-analysis including 15 studies and 8,745 patients with localized and metastatic BC, treated with adjuvant or neoadjuvant anthracyclines and/or trastuzumab [9]. We found that obesity/overweight was significantly associated with a higher risk of developing cardiotoxicity after anthracyclines and sequential anthracyclines and trastuzumab regimens. Associations were independent of study design, year of publication, drug regimen, and definitions of cardiotoxicity and overweight/obesity. However, we were unable to isolate the contribution of obesity-related cardiovascular risk factors, such as diabetes and hypertension, from that of obesity itself in this unadjusted analysis. The present study shows the association of obesity with cardiotoxicity in the early BC population. In this large scale, prospective, adjusted analysis, being obese was associated with a 3-fold increased risk of developing cardiotoxicity, regardless of other predictors of cardiotoxicity. Almost half of our patients had a  $BMI > 25 \text{ kg/m}^2$ , suggesting that overweight and obesity are commonly occurring risk factors for cardiotoxicity in patients with early BC.

Our results show that overweight and obese patients would certainly benefit from careful cardiac screening and follow-up during and after chemotherapy. Recently, several randomized studies showed a negative or modest effect of heart failure drugs such as beta blockers or angiotensin-renin system inhibitors in preventing cardiotoxicity [24]. In light of our results, we suggest that future studies should specifically focus on overweight and obese patients. Molecular pathways underlying the interaction between obesity and heart function, and leading to a higher probability of heart failure upon treatment with anthracyclines and trastuzumab, remain to be elucidated. The role of adipokines is considered to be at the intersection between obesity and CVDs, and adipokine dysregulation is a prominent hallmark of dysfunctional adipose tissue in obesity [25,26].

The current study has several limitations. One potential limitation is the relatively short follow-up of this interim analysis. With a mean follow-up of 22 months, we have likely captured the most likely timeframe for occurrence of early cardiotoxicity. In addition, centers measured LVEF using their preferred method, either echocardiography or MUGA. Nevertheless, variability could have been minimized if the same modality for repeated measurements had been required throughout the study. It was also not possible to study post-trastuzumab recovery in all patients due to missing data. Finally, many patients in the CANTO cohort had to be excluded due to lack of requisite baseline or follow-up LVEF measurements, leading to some selection bias.

#### Conclusion

In conclusion, obese women with early stage BC treated with anthracyclines and/or trastuzumab may be at higher risk of experiencing drops in LVEF. Our results should encourage further research into the mechanisms linking obesity to cardiotoxicity, and personalized risk assessment for cardiac complications following cardiotoxic therapies could offer several potential benefits to patients.

## **Supporting information**

**S1 Checklist. STROBE checklist of the OCATO study.** (DOCX)

#### Acknowledgments

The authors thank Isabel Grégoire, PhD, Medical Writer, Centre Georges-François Leclerc, for language editing.

#### **Author Contributions**

- **Conceptualization:** Charles Guenancia, Florence Dalenc, Philippe Rouanet, Antoine Arnaud, Olivier Arsene, Mahmoud Ibrahim, Johanna Wassermann, Fabrice André, Patrick Arveux.
- **Data curation:** Elisé G. Kaboré, Ines Vaz-Luis, Thierry Petit, Geneviève Boileau-Jolimoy, Jérôme Lemonnier, Patrick Arveux.

Formal analysis: Elisé G. Kaboré, Patrick Arveux.

- **Investigation:** Elisé G. Kaboré, Barbara Pistilli, Charles Coutant, Paul Cottu, Anne Lesur, Florence Dalenc, Olivier Arsene, Johanna Wassermann, Geneviève Boileau-Jolimoy, Fabrice André, Patrick Arveux.
- Methodology: Elisé G. Kaboré, Charles Guenancia, Antonio Di Meglio, Florence Dalenc, Mahmoud Ibrahim, Anne-Laure Martin, Fabrice André.
- Project administration: Ines Vaz-Luis, Charles Coutant, Anne-Laure Martin, Jérôme Lemonnier, Fabrice André.
- **Resources:** Antonio Di Meglio, Paul Cottu, Antoine Arnaud, Anne-Laure Martin, Patrick Arveux.
- Software: Anne-Laure Martin, Patrick Arveux.
- Supervision: Charles Guenancia, Barbara Pistilli, Charles Coutant, Paul Cottu, Anne Lesur, Thierry Petit, Philippe Rouanet, Mahmoud Ibrahim, Fabrice André, Patrick Arveux.
- Validation: Charles Guenancia, Ines Vaz-Luis, Barbara Pistilli, Charles Coutant, Paul Cottu, Anne Lesur, Thierry Petit, Florence Dalenc, Philippe Rouanet, Antoine Arnaud, Mahmoud Ibrahim, Johanna Wassermann, Geneviève Boileau-Jolimoy, Fabrice André, Patrick Arveux.
- Visualization: Philippe Rouanet, Olivier Arsene.
- Writing original draft: Elisé G. Kaboré.
- Writing review & editing: Elisé G. Kaboré, Charles Guenancia, Ines Vaz-Luis, Antonio Di Meglio, Barbara Pistilli, Charles Coutant, Paul Cottu, Anne Lesur, Thierry Petit, Florence Dalenc, Philippe Rouanet, Antoine Arnaud, Olivier Arsene, Mahmoud Ibrahim, Johanna Wassermann, Geneviève Boileau-Jolimoy, Anne-Laure Martin, Jérôme Lemonnier, Fabrice André, Patrick Arveux.

#### References

 Rosa GM, Gigli L, Tagliasacchi MI, Di Iorio C, Carbone F, Nencioni A, et al. Update on cardiotoxicity of anti-cancer treatments. Eur J Clin Invest. 2016; 46:264–84. https://doi.org/10.1111/eci.12589 PMID: 26728634

- Volkova M, Russell R. Anthracycline cardiotoxicity: prevalence, pathogenesis and treatment. Curr Cardiol Rev. 2011; 7:214–20. https://doi.org/10.2174/157340311799960645 PMID: 22758622
- Santoni M, Guerra F, Conti A, Lucarelli A, Rinaldi S, Belvederesi L, et al. Incidence and risk of cardiotoxicity in cancer patients treated with targeted therapies. Cancer Treat Rev. 2017; 59:123–31. https:// doi.org/10.1016/j.ctrv.2017.07.006 PMID: 28818671
- Koene RJ, Prizment AE, Blaes A, Konety SH. Shared risk factors in cardiovascular disease and cancer. Circulation. 2016; 133:1104–14. https://doi.org/10.1161/CIRCULATIONAHA.115.020406 PMID: 26976915
- Majed B, Moreau T, Asselain B, Curie Institute Breast Cancer Group. Overweight, obesity and breast cancer prognosis: optimal body size indicator cut-points. Breast Cancer Res Treat. 2009; 115:193–203. https://doi.org/10.1007/s10549-008-0065-7 PMID: 18546073
- Majed B, Moreau T, Senouci K, Salmon RJ, Fourquet A, Asselain B. Is obesity an independent prognosis factor in woman breast cancer? Breast Cancer Res Treat. 2008; 111:329–42. <u>https://doi.org/10.</u> 1007/s10549-007-9785-3 PMID: 17939036
- Calle EE, Kaaks R. Overweight, obesity and cancer: epidemiological evidence and proposed mechanisms. Nat Rev Cancer. 2004; 4:579–91. https://doi.org/10.1038/nrc1408 PMID: 15286738
- Renehan AG, Tyson M, Egger M, Heller RF, Zwahlen M. Body-mass index and incidence of cancer: a systematic review and meta-analysis of prospective observational studies. Lancet. 2008; 371:569–78. https://doi.org/10.1016/S0140-6736(08)60269-X PMID: 18280327
- Guenancia C, Lefebvre A, Cardinale D, Yu AF, Ladoire S, Ghiringhelli F, et al. Obesity as a risk factor for anthracyclines and trastuzumab cardiotoxicity in breast cancer: a systematic review and meta-analysis. J Clin Oncol. 2016; 34:3157–65. https://doi.org/10.1200/JCO.2016.67.4846 PMID: 27458291
- Mitra MS, Donthamsetty S, White B, Mehendale HM. High fat diet-fed obese rats are highly sensitive to doxorubicin-induced cardiotoxicity. Toxicol Appl Pharmacol. 2008; 231:413–22. https://doi.org/10.1016/ j.taap.2008.05.006 PMID: 18674790
- Guenancia C, Hachet O, Aboutabl M, Li N, Rigal E, Cottin Y, et al. Overweight in mice, induced by perinatal programming, exacerbates doxorubicin and trastuzumab cardiotoxicity. Cancer Chemother Pharmacol. 2016; 77:77–85. https://doi.org/10.1007/s00280-016-2995-9 PMID: 26914236
- Vaz-Luis I, Cottu P, Mesleard C, Martin AL, Dumas A, Dauchy S, et al. UNICANCER: French prospective cohort study of treatment-related chronic toxicity in women with localised breast cancer (CANTO). ESMO Open. 2019; 4:e000562. https://doi.org/10.1136/esmoopen-2019-000562 PMID: 31555487
- Zamorano JL, Lancellotti P, Rodriguez Muñoz D, Aboyans V, Asteggiano R, Galderisi M, et al. 2016 ESC position paper on cancer treatments and cardiovascular toxicity developed under the auspices of the ESC Committee for Practice Guidelines: the task force for cancer treatments and cardiovascular toxicity of the European Society of Cardiology (ESC). Eur Heart J. 2016; 37:2768–801. https://doi.org/ 10.1093/eurheartj/ehw211 PMID: 27567406
- Physical status: the use and interpretation of anthropometry. Report of a WHO Expert Committee. World Health Organ Tech Rep Ser. 1995; 854:1–452. PMID: 8594834
- Armenian S, Lacchetti C, Barac A, Carver J, Constine L, Denduluri N. Prevention and monitoring of cardiac dysfunction in survivors of adult cancers: American Society of Clinical Oncology clinical practice guideline. J Clin Oncol. 2017; 35:893–911. <u>https://doi.org/10.1200/JCO.2016.70.5400</u> PMID: 27918725
- McGowan JV, Chung R, Maulik A, Piotrowska I, Walker JM, Yellon DM. Anthracycline chemotherapy and cardiotoxicity. Cardiovasc Drugs Ther. 2017; 31:63–75. https://doi.org/10.1007/s10557-016-6711-0 PMID: 28185035
- Colombo A, Sandri MT, Salvatici M, Cipolla CM, Cardinale D. Cardiac complications of chemotherapy: role of biomarkers. Curr Treat Options Cardiovasc Med. 2014; 16:313. https://doi.org/10.1007/s11936-014-0313-6 PMID: 24771223
- Kongbundansuk S, Hundley WG. Noninvasive imaging of cardiovascular injury related to the treatment of cancer. JACC Cardiovasc Imaging. 2014; 7:824–38. <u>https://doi.org/10.1016/j.jcmg.2014.06.007</u> PMID: 25124015
- Agmon Nardi I, lakobishvili Z. Cardiovascular risk in cancer survivors. Curr Treat Options Cardiovasc Med. 2018; 20:47. https://doi.org/10.1007/s11936-018-0645-8 PMID: 29705862
- Linschoten M, Teske AJ, Cramer MJ, van der Wall E, Asselbergs FW. Chemotherapy-related cardiac dysfunction: a systematic review of genetic variants modulating individual risk. Circ Genom Precis Med. 2018; 11:e001753. https://doi.org/10.1161/CIRCGEN.117.001753 PMID: 29557343
- Kenchaiah S, Sesso HD, Gaziano JM. Body mass index and vigorous physical activity and the risk of heart failure among men. Circulation. 2009; 119:44–52. https://doi.org/10.1161/CIRCULATIONAHA. 108.807289 PMID: 19103991

- 22. Kenchaiah S, Chesebro JH. The epidemiologic association between obesity and heart failure. Am Coll Cardiol Ext Learn. 2017; 49:4–6. PMID: 29527015
- 23. Hirayama A, Goto T, Shimada YJ, Faridi MK, Camargo CA, Hasegawa K. Association of obesity with severity of heart failure exacerbation: a population-based study. J Am Heart Assoc. 2018; 7:e008243. https://doi.org/10.1161/JAHA.117.008243 PMID: 29545263
- Guglin M, Munster P, Fink A, Krischer J. Lisinopril or Coreg CR in reducing cardiotoxicity in women with breast cancer receiving trastuzumab: a rationale and design of a randomized clinical trial. Am Heart J. 2017; 188:87–92. https://doi.org/10.1016/j.ahj.2017.03.010 PMID: 28577685
- Koliaki C, Liatis S, Kokkinos A. Obesity and cardiovascular disease: revisiting an old relationship. Metabolism. 2019; 92:98–107. https://doi.org/10.1016/j.metabol.2018.10.011 PMID: 30399375
- Ouchi N, Parker JL, Lugus JJ, Walsh K. Adipokines in inflammation and metabolic disease. Nat Rev Immunol. 2011; 11:85–97. https://doi.org/10.1038/nri2921 PMID: 21252989